Author’s response to reviews

Title: Endoscopic Repair of Lateral Sphenoid Encephaloceles: A Case Series

Authors:
Mitchell Gore (mgoremdphd@gmail.com)

Version: 2 Date: 15 Nov 2017

Author’s response to reviews:
1. Please provide a section after References where you provide the figure legends for your figure files. For each figure, the following information should be provided:
   - Figure number
   - Short title of figure in sentence case (required)
   - Detailed legend (optional)

On page 7 line 1 I have added the figure legends section with the figure number and a short title for each figure.

Example:
Figure 3. Liver smear from a 6-week-old embryo (CR length 12 mm). A remarkable cluster of macrophages is seen among primitive erythroblasts (see also Fig. 4). A granuloblast is present at the upper right. H & E, ×1,150. (Scott 1974)

2. In the Consent for publication section, please indicate from whom permission was obtained, and whether it was written or verbal. If your manuscript contains any individual person’s data in any form (including individual details, images or videos), consent to publish must be obtained from that person. Please provide details of this consent in this section.

In the consent for publication section page 5 line 5 I have added the notation that written consent was obtained for all patients included in the study.
3. We noticed that the data provided in Table 1 contains potentially identifiable information, i.e. age, gender, others. It is BioMed Central policy to not publish more than 2 indirect identifiers without explicit consent for publication from the participants, as per this paper (http://trialsjournal.biomedcentral.com/articles/10.1186/1745-6215-11-9).

Please confirm that you have obtained consent to publish these individual details, and state this in the Consent for publication section. If you have not obtained consent to publish, then please amend the dataset to address this issue, by providing the ages as a range (i.e. 40-45).

Written consent to publish individual demographic details and operative images was obtained from all patients. In the consent to publish section page 5 line 5 I have noted that consent to publish demographic details and radiographic and intraoperative images was obtained from all patients.

For more information, please see our editorial policies: https://www.biomedcentral.com/getpublished/editorial-policies#consent+for+publication

4. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript

5. Page 5 line 10 I have noted that all data supporting the study is included within the manuscript and have noted this in the language/format noted in the instructions for authors.